Author's response to reviews

Title: How does gender influence the recognition of cardiovascular risk and adherence to self-care recommendation? A study in Polish primary care

Authors:

Ireneusz Szymczyk (rpmr@poczta.onet.pl)
Ewa Wojtyna (ewa.wojtyna@us.edu.pl)
Witold Lukas (rpmr@poczta.onet.pl)
Joanna Kępa (rpmr@poczta.onet.pl)
Teresa Pawlikowska (teresa.pawlikowska@warwick.ac.uk)

Version: 2 Date: 5 April 2013

Author's response to reviews:

Dear Editors,

As recommended by the Editors, the current version of the manuscript introduced a unique e-mail addresses of each of the authors.

Yours faithfully,
Authors